Re Manuscript NO: 70138
Dear editor,

Thank you for your letter and for the reviewers comments concerning our manuscript entitled “Endoscopic Surgery for Intraventricular Hemorrhage: A comparative study and Single Center Surgical Experience”, NO: 70138. Those suggestions are helpful. We have studied reviewers’ comments and make correction.

Answer to reviewer(s)’ commends:
Reviewer 1:
1, The cited references should be up to date within recent 5 years

We have updated the cited references according to your advice and rearrange them all.
Reviewer 2:

2, Please detail described any complications in this endoscopic procedure.

We have added some possible complications in this endoscopic procedure and analyze the superiority of our endoscopic procedure in discussion part.

Besides, and we have improved the quality of Figure1 and format the table in our paper, then prepare and arrange the figures using PowerPoint.

Finally, thank you for accepting our manuscript. If you have any other problems, please don't hesitate to contact us.

Sincerely yours,
Xiang.